Author's response to reviews

Title: Evaluation of Work Disability and the International Classification of Functioning, Disability and Health: What to Expect and What not

Authors:

Jessica Anner (janner@uhbs.ch)
Urban Schwegler (urban.schwegler@paranet.ch)
Regina Kunz (rkunz@uhbs.ch)
Bruno Trezzini (bruno.trezzini@paranet.ch)
Wout De Boer (wdeboer@uhbs.ch)

Version: 2 Date: 20 January 2012

Author's response to reviews:

Dear Editor,

We are happy to submit the revision to BMC Public Health with emphasis on disability evaluation and the International Classification of Functioning, Disability and Health. The article changed from a research article to a debate one.

The changes we made for the debate article:

- Added keywords
- Removed methods part
- Rephrased the text in debate terminology rather than in empirical terms

Sincerely,
Jessica Anner